Author's response to reviews

**Title:** Population awareness of risks related to medicinal product use in Vientiane Capital, Lao PDR: a cross-sectional study for public health improvement in low and middle income countries

**Authors:**

- Céline Caillet (celine.caillet@univ-tlse3.fr)
- Chanvilay Sichanh (singsing_chanvilay@yahoo.fr)
- Lamphone Syhakhang (syhakhangl@yahoo.com)
- Cyrille Delpierre (cyrildelpierre@yahoo.fr)
- Chanthanom Manithip (manithipchanthanom@gmail.com)
- Mayfong Mayxay (mayfong@tropmedres.ac)
- Maryse Lapeyre-Mestre (maryse.lapeyre-mestre@univ-tlse3.fr)
- Paul N Newton (paul@tropmedres.ac)
- Anne Roussin (anne.roussin@univ-tlse3.fr)

**Version:** 2

**Date:** 28 October 2014

**Author's response to reviews:**

'Ethical approval', and 'Abbreviations' sections were added to the manuscript. In addition, the 'Authors' Contribution' section was detailed by using initials to refer to each author's contribution.